Author's response to reviews

Title: Family-based intervention in adolescent restrictive eating disorders: Early treatment response and low weight suppression is associated with favourable one-year outcome

Authors:

Ingemar Swenne (ingemar.swenne@kbh.uu.se)
Thomas Parling (thomas.parling@psyk.uu.se)
Helena Salonen Ros (helena.salaonen.ros@akademiska.se)

Version: 3 Date: 17 Aug 2017

Author’s response to reviews:

Family-based intervention in adolescent eating disorders: Early treatment response and low weight suppression predicts favourable one-year outcome

I Swenne, T Parling, H Salonen Ros

BPSY-D-17-00183

Response to Reviewer 2

*The title has been changed in the manuscript submission system.

*The conclusion section of the Abstract has been changed in line with changes in the Discussion, i.e. “prediction” has been changed to “association”.

*The typo on page 6 has been corrected.

*The cut-off for the weight criterion has been added to the “Participants” section. As a consequence there was a minor rearrangement of references.

*Ranges have been added to Table 1.